Author's response to reviews

Title: Promoter methylation of CDKN2A and the lack of p16 expression characterize patients with hepatocellular carcinoma

Authors:

Antal Csepregi (csepregia@yahoo.de)
Matthias PA Ebert (ebertm@gmx.de)
Christoph Röcken (Christoph.Roecken@uh-sk.de)
Regine Schneider-Stock (regine.schneider-stock@med.ovgu.de)
Juliane Hoffmann (juliane.hoffmann@med.ovgu.de)
Hans-Ulrich Schulz (hans-ulrich.schulz@med.ovgu.de)
Albert Roessner (albert.roessner@med.ovgu.de)
Peter Malfertheiner (peter.malfertheiner@med.ovgu.de)

Version: 7 Date: 19 June 2010

Author's response to reviews: see over
To
Rachel Neilan, MSc
Assistant Scientific Editor
BMC Cancer
BioMed Central
Editorial Board

Title: Promoter Methylation of CDKN2A and Lack of p16 Expression Characterize Patients With Hepatocellular Carcinoma.

MS: 1344631501293011

Authors: Antal Csepregi, MD, PhD, Matthias P.A. Ebert, MD, Christoph Röcken, MD, Regine Schneider-Stock, PhD, Juliane Hoffmann, PhD, Hans-Ulrich Schulz, MD, Albert Roessner, MD, Peter Malfertheiner, MD

Dear Mrs Neilan,

The following changes have been made in the manuscript:

Major revisions:

1. Tables were all edited. Data on tumor differentiation grade and AFP must have been excluded.
2. Figures are cited in consecutive order and Figure 3 is also cited in the text.
3. All figures and the manuscript have been submitted again.

The invoice about article processing charge + administrative charge (EUR 1200 + VAT and £50 + VAT) must be to:

Uniklinikum Magdeburg
Otto-von-Guericke University
Leipziger Straße 44
D-39120 Magdeburg

Tax number: DE 811640443

The invoice must be sent to:
Prof. Dr. med. Peter Malfertheiner
Department of Gastroenterology, Hepatology and Infektiology
Otto-von-Guericke University
Leipziger Straße 44
D-39120 Magdeburg
With kind regards,

Antal Csepregi, MD, PhD